Author's response to reviews

Title: Predictors of labor abnormalities in university hospital: A case control study

Authors:

Wayou Abraham (wayouabrham@gmail.com)
Yifru Berhan (yifrub@yahoo.com)

Version: 2 Date: 18 September 2013

Author's response to reviews:

Dear BMC pregnancy and childbirth editor,

I am submitting a research article entitled “PREDICTORS OF LABOR ABNORMALITIES IN UNIVERSITY HOSPITAL: A CASE CONTROL STUDY” for possible publication. We are interested in your journal because it is open access and highly visible. This manuscript was extracted from MSC thesis and it talks about one of the common obstetric problems but with very limited data from the third world where the magnitude of the problem is overwhelming. I would like to assure you that the manuscript was not submitted in part or as a whole to another journal.

Thank you

Yifru Berhan M.D

Sept /2013